Author’s response to reviews

Title: Age and manifestation related symptoms in familial adenomatous polyposis

Authors:

Roland S. Croner (Roland.Croner@chir.imed.uni-erlangen.de)
Wolfgang M. Bruckl (wbrueckl@molmed.uni-erlangen.de)
Bertram Reingruber (Bertram.Reingruber@chir.imed.uni-erlangen.de)
Werner Hohenberger (Werner.Hohenberger@chir.imed.uni-erlangen.de)
Klaus Guenther (Klaus.Guenther@chir.imed.uni-erlangen.de)

Version: 9 Date: 25 February 2005

Author’s response to reviews:

To whom it may concern,
1. Section headings: sentence case for all headings was used
2. References: The reference list was formatted in the journal’s style
3. Tables: They were oriented in portrait rather than in landscape.
4. Figures: figures were cropped as closely as possible to minimise white space around the image

Sincerely
RS. Croner